

JGR Earth Surface

RESEARCH ARTICLE

10.1029/2023JF007297

Key Points:

- Mid-Holocene advance of the Quelccaya Ice Cap (QIC) is documented using ¹⁴C dating of in situ vegetation exposed during the recent retreat
- Plant ages of ~4.5 to ~7.1 ka BP provide evidence of a mid-Holocene climate transition and the QIC ice-margin expansion
- Analysis of satellite imagery from 1985 to 2020 documents a recent rapid retreat (~20 m/yr) of the QIC margin

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to: K. Lamantia,

lamantia.31@osu.edu

Citation:

Lamantia, K., Thompson, L., Davis, M., Mosley-Thompson, E., & Stahl, H. (2023). Unique collections of ¹⁴C-dated vegetation reveal Mid-Holocene fluctuations of the Quelccaya Ice Cap, Peru. Journal of Geophysical Research: Earth Surface, 128, e2023JF007297. https://doi.org/10.1029/2023JF007297

Received 12 JUN 2023 Accepted 6 NOV 2023

Author Contributions:

Conceptualization: Kara Lamantia, Lonnie Thompson, Henry Stahl Data curation: Kara Lamantia, Henry Stahl Funding acquisition: Lonnie Thompson Methodology: Kara Lamantia Resources: Lonnie Thompson Software: Kara Lamantia Supervision: Lonnie Thompson, Ellen Mosley-Thompson Visualization: Kara Lamantia, Mary Davis Writing – original draft: Kara Lamantia

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Unique Collections of ¹⁴C-Dated Vegetation Reveal Mid-Holocene Fluctuations of the Quelccaya Ice Cap, Peru

Kara Lamantia^{1,2} , Lonnie Thompson^{1,2} , Mary Davis¹ , Ellen Mosley-Thompson^{1,3} , and Henry Stahl²

¹Byrd Polar and Climate Research Center, Ohio State University, Columbus, OH, USA, ²School of Earth Sciences, Ohio State University, Columbus, OH, USA, ³Department of Geography, Ohio State University, Columbus, OH, USA

Abstract Several studies have analyzed the ice margin behavior of the Quelccaya Ice Cap (QIC), Earth's largest tropical ice cap, through the Holocene. However, continuous integration of new information to produce a more cohesive history of the QIC is necessary. Here, the radiocarbon dates of 33 rooted plant specimens collected in situ along the western ice margin between 2002 and 2018 reveal the timing of its past extent as it advanced during the mid-Holocene. The most recent evaluation of collected specimens indicates that the QIC margin advanced ~350 m down the Challpacocha Valley between 7.1 and 4.5 ka BP. Past studies of documented ice extent on the western side of the QIC based on a variety of techniques are compiled to create a more comprehensive history of the QIC's behavior throughout the Holocene. Records of documented ice extent, as well as other proxy records, indicate a climate transition ~5–7 ka BP that created the proper environmental conditions for the expansion of the QIC. Evidence from nearby valleys indicates that the QIC behaved similarly to the documented ice extent in the Challpacocha Valley in response to Holocene climatic fluctuations. The ability to collect the plant specimens and recent analysis of satellite imagery reveals rapid retreat rates of the western outlet glaciers from 1985 to 2020, leaving the western margin of the QIC at its smallest extent since the mid-Holocene.

Plain Language Summary The area around the Quelccaya Ice Cap (QIC) is unique as wetland plant samples have been found preserved in their growth positions, buried by past advancing ice, whereas the ice typically removes vegetation and soil in its advance down a valley. Using the ¹⁴C dates of these plants, we map a 350-m advance of the ice margin between 7.1 and 4.5 ka BP. Combined with a variety of previous studies involving different methods and records from ice and lacustrine sediment cores, a larger picture of the QIC's behavior comes into view. A largely accepted global climate shift between ~7 and 5 ka BP occurred in tandem with the QIC's advance. Although the QIC has fluctuated before, at no time has the western margin receded to the current extent since the mid-Holocene. This is indicated by the exposure of the plants and current margin retreat rates calculated from satellite imagery (1985–2020).

1. Introduction

The high Peruvian Andes contain 70% of Earth's tropical glaciers (Kaser & Osmaston, 2002), which provide a seasonal water resource for consumption, hydroelectric power, and agriculture. Unfortunately, much of the ice in these mountains has been retreating in recent decades (Bradley et al., 2006). Earth's largest tropical ice cap, the Quelccaya Ice Cap (QIC), is located in the southeastern Peruvian Andes (Figure 1). Recent satellite observations indicate a 46% loss in Quelccaya's surface area between 1976 and 2020 (Thompson et al., 2021). The retreating ice has created a need for an improved understanding of the QIC's past behavior to better anticipate its future.

Numerous studies of the QIC detail a variety of methods that have been utilized to investigate the ice cap's past extent. These include radiocarbon dating of organic material found in sediment (Kelly et al., 2012), radiocarbon dating of vegetation remains (Buffen et al., 2009), cosmogenic nuclide surface exposure age dating (¹⁰Be) (Kelly et al., 2015; Stroup et al., 2014, 2015), lake sediment records (Rodbell et al., 2008; Stroup et al., 2015), and paleo-glacier reconstruction from moraine mapping (Mark et al., 2002; Mercer & Palacios, 1977). While many studies provide regional paleoclimate records from ice cores, moraines, and lake sediments (Rodbell et al., 2008; Stroup et al., 2015; Thompson, 2000), there are few precisely dated glacial landforms in the tropics (Kelly et al., 2015) which limits the understanding of glacier and ice cap behavior in response to changes in climate.





Writing – review & editing: Kara Lamantia, Lonnie Thompson, Mary Davis, Ellen Mosley-Thompson The mapping and dating of moraines (Figure 1) has provided valuable but discontinuous records of past glacial fluctuations (Stroup et al., 2015). However, field observations from 1976 to 2020 confirm the 46% decrease in the QIC's surface area over 45 years is largely attributed to increasing air temperature (Thompson et al., 2021; Yarleque et al., 2018). During several expeditions to the QIC from 2002 through 2007, dozens of well-preserved in situ plant specimens were collected along a recently deglaciated ice margin (Buffen et al., 2009) with additional plant specimens subsequently collected through 2018. Radiocarbon dating of these plants, which were exposed by the recent glacier retreat, provides ages of their in situ burial by the advancing ice margin (Anderson et al., 2008; Thompson et al., 2006). While the preservation of in situ plants is extremely rare, their existence provides insight into the likelihood of a time when the QIC was frozen to the bedrock and thus experienced minimal basal sliding and meltwater erosion. In this study, we use the unique radiocarbon-dated plant samples found along the North Lake Lobe on the western margin of the QIC to analyze the ice cap's extent in the mid-Holocene and compare it to other proxy histories from the QIC and the surrounding area. Past studies involving a variety of methods and locations are combined with our data to create a more comprehensive understanding of the QIC's past advances and retreats. Today, the addition of available remotely sensed satellite observations allows us to calculate the rate of its present-day retreat.

2. Study Area

2.1. Geography and Geology

The QIC ($13^{\circ}56'S$; $70^{\circ}50'W$) is located in the Cordillera Vilcanota on the northern edge of the Peruvian/Bolivian Altiplano, a large plateau in the central Andes. The average elevation of the Altiplano is ~3,750 m above sea level (m a.s.l.) with an area of ~200,000 km². The summit elevation of the highest of the QIC's four domes is 5,670 m a.s.l., and short, steep outlet glaciers descend to elevations as low as 4,950 m a.s.l. on the escarpment of the underlying ignimbrite plateau, an igneous, primarily pumice, pyroclastic flow deposit (Mercer & Palacios, 1977). Landsat-9 satellite imagery from August 2022 indicates that the southern outlet glaciers do not extend below 5,200 m a.s.l. while the terminus of Qori Kalis lies at approximately 5,000 m a.s.l. The low slope of the QIC leaves it vulnerable to changes in the mean elevation of the 0°C isotherm (Thompson et al., 2017). The North Lake, where the plant samples were collected, is at the terminus of the North Lake Lobe of the QIC (Figure 1).

2.2. Regional Climate

The QIC experiences a small range in seasonal temperatures and solar radiation reaching the surface, which varies less than 30% from winter to summer with the maximum insolation occurring a month or two before the rainy season (Garreaud et al., 2003). Seasonal precipitation variations are well defined and mostly occur during the warmer wet season (austral summer) between December and February (Garreaud et al., 2003). This moisture originates in the Atlantic Ocean, and is transported over the Amazon Basin by northeasterly trade winds before it is deposited on the Altiplano (Hurley et al., 2015; Vuille et al., 2000). The Andes experienced an increase in temperature of 0.39°C per decade between 1951 and 1999, which is consistent with global climate trends (Vuille & Bradley, 2000). Recent studies (Bradley et al., 2006; Pepin, 2015; Vuille et al., 2015) document increased warming in the high elevation mountain regions with the high Andes maximum and mean temperatures continuing to increase (Chimborazo, 2022; Toledo et al., 2022).

3. Methods

3.1. Sample Collection

The in situ vegetation samples, which were identified as *Distichia muscoides* (Juncaceae), were collected between 2002 and 2018 from the margin of the North Lake Lobe (Table S1 in Supporting Information S1). *Distichia* is the predominant species among the 10 common wetland plant species in the region. It is a diecious, cushion-forming plant that is adapted to the diurnal freeze-thaw cycles of the Altiplano's climate (Buffen et al., 2009). The plants were collected from their growth positions around the edges of the lakes and within the banks of fine-grained sediment as well as along the ice front (Photo S1 in Supporting Information S1). The plant remains were found only near the lake edge, as *Distichia* is present in bofedale communities, which are peat-forming, high-altitude tropical and subtropic wetlands within the Andes (Gould et al., 2010). The plants indicates that



Journal of Geophysical Research: Earth Surface





Figure 1. The Quelccaya Ice Cap with all noted sample locations, dated moraines, and analyzed valleys.

a bofedale environment once existed around the lake margin. The collected plant samples were found largely clustered around the northern and northeastern margins of the lake nearest to the retreating ice margin. The ignimbrite bedrock around the North Lake consists of poorly sorted silica-based volcanic ash, which prevents the influx of present-day carbon from soils or peats. No vegetation currently grows in the collection area, and only sparse patches of early colonizing vegetation are present in the immediate vicinity (Gould et al., 2010), confirming the lack of potential influx of present-day carbon. A total of 62 plants were collected over the 16-year period, consisting of 51 surface samples and 11 subsurface samples down to depths of 100 cm. When possible, subsurface samples were collected from layers of sediment deposited in various areas around the lake and their collection depths were measured as the distance below the ground surface. As noted in Buffen et al. (2009), the dominant top layer of sediment is composed of a massive diamict characterized by large subangular and subrounded cobbles supported by a matrix of clay and silt. Location coordinates and elevation for each sample were recorded using Garmin Global Positioning System (GPS) handheld units.

3.2. Radiocarbon Dating

All samples were dated using AMS radiocarbon techniques at the Woods Hole Oceanographic Institution's National Ocean Science Accelerator Mass Spectrometry Facility (NOSAMS) and the Center for Accelerator Mass Spectrometry (CAMS) at the Lawrence Livermore National Laboratory (Buffen et al., 2009). If a sample contained sufficient organic material, it was dated multiple times at one or both facilities. The ¹⁴C dates in Buffen



Journal of Geophysical Research: Earth Surface



Figure 2. View of the North Lake Lobe, collected plant sample locations, and isochron contours created to approximate ice extent over time as depicted on WorldView-3 (WV03) satellite imagery with 1.24-m multispectral resolution. Isochron contours were extrapolated from the sample points according to Section 3.3.

et al. (2009) were calibrated using the Calib 5.0.1 radiocarbon calibration program with the Southern Hemisphere data set SHCal04 (McCormac et al., 2004; Stuiver & Reimer, 1993) (Table S1 in Supporting Information S1). Composite ages were calculated for samples from the North Lake location using the "C_Combine" function of the OxCal4 "R_Combine" function and were weighted as a single sample for the composite site (Bronk Ramsey, 1995; McCormac et al., 2004). In cases where samples were radiocarbon dated more than once, the mean ages were calculated. Samples collected after 2009 were dated using the same procedures as those collected earlier. For this work, the radiocarbon dates were recalibrated using Calib Rev 8.2.0 with the Southern Hemisphere data set SHCal20 (McCormac et al., 2004).

3.3. Data Analysis

Among the 51 surface samples collected between 2002 and 2018, several groups were collected at the same location, and their mean radiocarbon ages were calculated, yielding 33 sample locations (Figure 2). Spatial interpolations were conducted on the age data points including Natural Neighbor, Kriging, and Spline. These interpolations were not assumed to be accurate due to the low number of data points (33) and the unknown topography of the lakebed around the glacial margin. Instead, they were used as an aid, along with the surrounding topography and notable glacial landforms, to generate ice burial isochrons from the data as a representative of past ice extent. The isochrons were set at intervals of 400 years to encompass all the data points between ~4.5 and ~7.1 ka BP. As the plants were not collected within the lake, interpolations of the ice boundary across the lake surface were necessary. Areas of high and low confidence were noted along each isochron, with high-confidence areas based on the spatial interpolation variances which do not extend more than 100 m from a singular data point



21699011, 2023, 11, Down

(Figure 2). Although of low confidence, the 7 ka BP isochron is unaligned with the other isochrons due to the flat topography near the current ice-edge. The 11 subsurface samples, the ages of which increase with depth, occurred in a stratigraphic sequence beneath the surface plants. The subsurface samples were buried by other processes at the glacial margin, such as moraine or meltwater movement and thus were not used for interpretation of the ice boundary extent.

Additionally, the North Lake Lobe ice margin was monitored for ice margin retreat rates using the Google Earth Engine Digitization Tool (GEEDiT) (Lea, 2018). The North Lake Lobe was observed via satellite imagery obtained from ASTER, Landsat 5, and Landsat 8 satellites (Table S3 in Supporting Information S1) and the ice margin was delineated every 5 years between 1985 and 2020, a feature of the GEEDiT tool. Due to the irregular shape of the ice margin, retreat rates during the 20th century were determined by drawing 10 transects between the oldest and youngest ice boundary lines to determine the mean value of ice margin movement over time.

4. Results

4.1. North Lake Lobe

Recalibration of the plant ¹⁴C ages produced a range of ~4.5 ka BP to ~7.1 ka BP (Table S1 in Supporting Information S1). The recalibration tool provides an integer age for each collected sample (e.g., 5,174 years) although the original data were provided with a margin of error which did not exceed \pm 40 years. Mean sample ages for groups collected from the same location were calculated as they were within 30 years of each other (Table S2 in Supporting Information S1). All the calculated ages of the remains decrease with distance from the ice margin moving northwest across the lake and down the Challpacocha Valley. Over a period of approximately 2,600 years, the ice advanced ~350 m down the valley to the northwest, and no additional data were found to indicate a larger ice extent (Figure 2). While the plant locations provide evidence for the extent of the ice coverage, the rate of advance is not assumed to be a single continuous process. Instead, the interbedded diamict and stratified sediment stratigraphic sequence described by Buffen et al. (2009) indicates a variably advancing ice margin. During this time the ice margin would have entirely covered the present-day lake. The location of the plants around the lake margin provides an indication of the mid-Holocene behavior of the North Lake Lobe and possibly the QIC as a whole.

Analysis of the North Lake Lobe using the GEEDiT tool shows a retreat rate of ~14 m/yr over a 35-year period between 1985 and 2020 (Figure 3). The 2020 margin was re-traced and edited using WV03 imagery for better resolution. Recent ice margin retreat rates for the Qori Kalis glacier (~28 m/yr) (Figure S1 in Supporting Information S1) and the Huancané Valley glacier (~18 m/yr) (Figure S2 in Supporting Information S1) were calculated using the GEEDiT tool and their retreat rates were similar to that of the North Lake Lobe.

5. Discussion

5.1. Integrating Past Studies

The margins of Quelccaya and its several outlet glaciers have been studied for decades to reconstruct past ice extent, understand the behavior of the ice cap, and determine the dominant drivers of its changing ice extent. To understand the larger picture beyond the North Lake Lobe's behavior and assess surface area changes of the QIC, data from past studies were obtained to augment the North Lake data (all shown in Table S1 of the Supporting Information S1) and improve the overall understanding of the QIC's past ice extent. Figure 4 presents a graphical summary of the movement of the Challpacocha Valley ice margin inferred from the paleo-data points and satellite imagery (Table S1 in Supporting Information S1) relevant to the 1985 position derived from satellite observations.

5.1.1. Challpacocha Valley

The North Lake Lobe descends into the Challpacocha Valley (Figure 1) where the dating of moraines and lacustrine sediment cores were reported in previous studies (Mercer & Palacios, 1977; Stroup et al., 2015). Radiocarbon-dated peat samples from two end moraine belts in the valley (Huancané I and II) have ages ranging from 12.3 to 11.2 ka BP within the Huancané II moraine (located ~4 km from the summit of the QIC) to ~900 yr BP within the Huancané I moraine (located ~1 km from the summit) (Mercer & Palacios, 1977). Recalibration



Journal of Geophysical Research: Earth Surface

10.1029/2023JF007297



Figure 3. Recent retreat of the North Lake Lobe as outlined using satellite imagery and the GEEDiT Tool (Lea, 2018).







of these ¹⁴C ages resulted in an age range of 14.1 to 13 ka BP for the Huancané II moraine and 760 yr BP for the Huancané I moraine. The addition of the Mid-Holocene plant sample data provides an indication of a retreat further up the valley than Little Ice Age (LIA) dated Huancané I moraine position. This indicates that a period of retreat must have occurred between the youngest Huancané II moraine (~13 ka BP) and the oldest plant (~7 ka BP), leading to the conclusion that the QIC at was least small as it is today. Additional evidence from organic material within moraines in the upper portion of the Challpacocha Valley reveals that the QIC expanded ~800 yr BP and reached a maximum extent at ~300 yr BP, which is consistent with ages (270 ± 80 yr BP) from the uppermost portion of Huancané I (Mercer & Palacios, 1977) and with the small increase in clastic sediment flux evident in the Lake Pacococha record (Rodbell et al., 2008). While the dates from the moraines provide limits on the glaciation, they do not provide information about ice margin fluctuations between those limits.

The presence of *D. muscoides* provides evidence that the climate was warmer and drier in this region at ~6 ka BP and that until recently ice margin retreat had not occurred since the burial of the plants (Figure 4) (Thompson et al., 2006, 2013). The mid-Holocene ¹⁴C ages of the plants confirm that this area adjacent to the QIC was not covered by ice at that time and their excellent preservation indicates that the QIC margin expanded beyond the current extent shortly after plant death. The burial of these soft-bodied plants by the ice coincided with the abrupt onset of a wet, cool period that led to the expansion of the QIC. Evidence for the QIC ice advance during the mid-Holocene was deduced by Rodbell et al. (2008) from studies of clastic sediment flux in nearby Lake Pacococha, which receives meltwater directly from the QIC. Clastic sediment flux abruptly increased around 5 ka BP and peaked ~4 ka BP, after which it declined to present values with a small increase during the Little Ice Age (LIA). The abrupt increase in clastic sediment flux corresponds closely to the burial of the plants by the advancing ice margin ~5,135 ± 45 cal yr BP.

The unique collection of the *Distichia* plants provides an innovative way to assess glacial margin movement over the course of the mid-Holocene. As radiocarbon dates from moraines typically offer a snapshot of the glacial margin during a point in time, the distance between each moraine cannot be used to draw any conclusions about the intervening nature of the ice movement. Due to both the minimal physical distance and radiocarbon ages between the *Distichia* plants, it becomes possible to track the advance of the glacial margin at the North Lake, filling in the lack of information about the distinct behavior of the glacial margin during the mid-Holocene. The advancing QIC margin that buried plants in their growth positions raises further discussion regarding the basal ice being frozen to the bedrock and thus enabling the preservation of the plants by reducing the flow of meltwater and basal erosion processes. Rodbell et al. (2008) note that tropical glaciers tend to have steep mass balance gradients and are likely to respond to climatic change with little lag time.

Additionally, the recent exposure of the plants coupled with satellite imagery enable the analysis of the present-day retreat at the North Lake as well as multiple locations across the QIC. The North Lake retreat rate from 1985 to 2020 (~14 m/yr) is compatible with that of the Qori Kalis glacier retreat (~28 m/yr) and the retreat at the Huancané Valley (~18 m/yr). A 46% loss of surface area was observed via satellite imagery over the entire QIC between 1976 and 2020 (Thompson et al., 2021). Other studies cite similar evidence for the recent retreat of glaciers across the region such as the Ampay glacier from 1991 to 2017, that experienced an area loss of 48% (Soto Carrión et al., 2022). The total glacial area of the Cordillera Blanca, Peru has declined by more than 30% from 1930 to present-day (Schauwecker et al., 2014) and the Cordillera Huaytapallana, Peru has lost 55% of its glacial surface coverage from 1984 to 2011 (López-Moreno et al., 2014). Similarly in Bolivia, a 30% area loss of glaciers in the Cordillera Real and Tres Cruces is observed from 2000 to 2016 (Seehaus et al., 2020). While discussions of the driving forces for these retreating margins in each respective locations vary, it is clear that the QIC and its outlet glaciers are presently retreating at an accelerated rate, along with many of the glaciers throughout the Andes.

5.1.2. Qori Kalis Glacier

Additional data from studies conducted around the QIC document the extent of other portions of the ice cap beyond the Challpacocha Valley during the mid-Holocene. Beryllium-10 (10 Be) ages from moraines around the Qori Kalis Lake (located ~1 km north of the North Lake) range between 1400 and 1800 CE (150 and 550 yr BP) (Stroup et al., 2014), indicating an ice retreat of 400 m over 400 years. Stroup et al. (2015) compared the clastic sediment flux record using sediment cores and 10 Be dating of boulders in the context of QIC behavior in the late Holocene. 10 Be ages were not incorporated with the radiocarbon dates because Kelly et al. (2015) indicated that as erosion is assumed to occur on the boulder surface, the ages are influenced by 10 Be inherited from

prior periods of exposure. It is noted that the Huancané I moraine can be traced almost continuously into the Qori Kalis Valley and is assumed to be apparently old due to prior exposure (Stroup et al., 2015). They are not intended for use as accurate exposure ages, but instead are used to examine the age differences between adjacent and nearby surfaces. However, the CH-2-Unit-2 record contains clastic rich (~95%) sediment, which suggests higher meltwater production and an increase in subglacial erosion, corresponding to the beryllium-indicated QIC retreat from 1490 to 1710 CE (Stroup et al., 2015). The monitoring of Qori Kalis has been discussed in several studies (Brecher & Thompson, 1993; Hanshaw & Bookhagen, 2014; Kelly et al., 2015; Lamantia, 2018; Stroup et al., 2014; Thompson et al., 2006). Qori Kalis has been monitored by aerial photography (Brecher & Thompson, 1993) beginning in 1963, by terrestrial photogrammetry from 1978 to 2004, and by stereo analysis from 2004 to 2017. Between 1963–1978 and 2004–2017, the overall retreat rate increased by an order of magnitude. The increase in the retreat rate between 1978 and 2017 was most likely impacted by the slope of the bedrock or the retreat of the terminus of the glacier from the growing proglacial lake.

The accelerating ice retreat on Qori Kalis is consistent with the retreat of other glaciers in the Cordillera Blanca and is characteristic of most low- and mid-latitude ice fields (Georges, 2004; Schauwecker et al., 2014; Thompson et al., 2006, 2021) and is similar to the shrinkage of the North Lake Lobe and of the QIC as a whole. As of 2020, the ice margin had retreated 200 m east of the lake that is located beyond the terminus of Qori Kalis (Figure S1 in Supporting Information S1). While a slow rate of advance during cooling and rapid retreat rates during warming is expected, as demonstrated by Brook and Buizert (2018), evidence from the North Lake Lobe and surrounding areas indicates accelerated recent retreat rates. While this type of retreat may have occurred during earlier periods of the Holocene, the data are not sufficiently robust to warrant a direct comparison with the recent higher temporal resolution data set.

5.1.3. Huancané Valley

The Huancané Valley, situated approximately 3 km south of the North Lake (Figure 1), has been investigated for indicators of the QIC's past extent. Radiocarbon dating of wetland plants growing up-valley from the Huancané II moraines provides their maximum and minimum limiting ages (Kelly et al., 2012). These results are compatible with ages near the current ice cap margin, which indicate that from \sim 7 to 5 ka BP the QIC was as small as it is today (Buffen et al., 2009; Thompson et al., 2013). Additionally, the Huancané II moraines outline an almost continuous paleo-ice margin ranging in age from ~12.4 to 11.2 ka BP, providing limiting ages for the extent of the ice margin in the Huancané valley (Kelly et al., 2015). The Huancané II extent was likely short-lived, as indicated by the lower flux of clastic sediment in Lake Pacococha that was dammed by the Huancané II moraines (Rodbell et al., 2008) originally mapped by Mercer and Palacios (1977). The Huancané I moraines indicate the extent of the ice margin further up the Huancané Valley at approximately 300 yr BP (Kelly et al., 2012; Mercer & Palacios, 1977; Stroup et al., 2014). The OIC has been retreating since that time, with marked acceleration within the last 50 years (Kelly et al., 2015). Previous monitoring of the Huancané Valley ice margin between 2000 and 2016 revealed a decrease in surface area that was ~28% greater during 1 year (2015-2016) than the mean annual rate of decrease during the previous 15 years (Thompson et al., 2017). Note that 2015–2016 was one of the strongest El Niño events in the last 100 years. Calculations from satellite imagery of the Huancané Valley from 1985 to 2020 using the GEEDiT Tool (Lea, 2018) indicate a ~18 m/year rate of retreat of the ice margin up the valley similar to its neighboring glaciers.

5.2. QIC Mid-Holocene Ice Advance as Part of a Global Climate Transition

The ages of the *Distichia* plants collected around the North Lake support a large body of evidence for the transition between ~5 and 7 ka BP from a warm Early Holocene to a cooler Late Holocene climate, not only in the central Andes but also on a global scale. Paleoclimate records from the tropics, including the Huascarán, Peru (Thompson et al., 1995) and Kilimanjaro (Thompson et al., 2002) ice core δ^{18} O time series, changes in Lake Titicaca water levels, water chemistry, and sediments (Baker et al., 2001, 2005), and pollen records from Peruvian and Bolivian lake cores (Abbott et al., 1996; Baker et al., 2009; Rodbell et al., 2022; Urrego et al., 2010; Weng et al., 2006) all suggest a cooling trend that started ~5.5 ka BP. Climate records from other locations in South America such as north central Chile (Tiner et al., 2018) and Columbia (Munoz Uribe, 2012), show that this middle Holocene transition extended beyond the central Andes. Lake sediments from Ecuador also indicate an insolation-linked decrease in the frequency of El Niño Southern Oscillation (ENSO) events (Mark et al., 2022; Moy et al., 2002; Rodbell et al., 1999). Climate records from extratropical locations also show that this cooling



21699011, 2023, 11, Downloa

from

com/doi/10.1029/2023JF007297 by Ohio State

University Ohio Sta,

Wiley Online Library on [27/11/2023].

. See the

event extended beyond the tropical regions. Evidence from peat cores in Patagonia (Huber & Markgraf, 2003), high resolution multi-proxy records from a Korean peninsula (Park et al., 2019), sediment and pollen records from Lake Constance water levels (Magny et al., 2006), North Atlantic marine sediment cores (Bond et al., 2001; Oppo & Cullen, 2003), and speleothems in Israel (Bar-Matthews et al., 1999) suggest cooler and wetter conditions at ~5 ka BP. These collective observations, which include the ages of the QIC plant samples, support the hypothesis of a large and rather abrupt mid-Holocene climatic shift (Figure S3 in Supporting Information S1).

Larger scale climate reconstructions focused on global mean surface temperature (GMST) indicate early Holocene warming (11.3 ka BP) leading to a temperature plateau between 9.5 and 5.5 ka BP and a long-term cooling of 0.7° C from 5.5 to ~0.1 ka BP (Marcott et al., 2013) as well as the peak warm period occurring ~6.5 ka BP followed by a steady cooling of -0.08° C/ky (Kaufman et al., 2020). Both the Marcott and Kaufman reconstructions indicate a period of global cooling, consistent with past paleotemperature/paleoclimate studies, which would create a favorable climate for the down-valley advance of the QIC. Other studies indicate early Holocene warmth between 40°N and 40°S, with models demonstrating steadily rising temperatures due to ice melt and greenhouse gas increases (Bova et al., 2021) followed by a mid-Holocene warming trend with associated cooling in the Northern Hemisphere (30°–90°N) (Liu et al., 2014). Bova et al. (2021) also noted that previous global reconstructions likely indicate that seasonal rather than annual temperature affected the outcome of previous global reconstructions and the discrepancy between proxy records and the model-data could be attributed to either seasonal bias in the sea surface temperature reconstructions or the model bias in regional climate sensitivity (Liu et al., 2014). Most recently, evidence from proxy records and climate model simulations indicates a mild-millennial scale global thermal maximum during the mid-Holocene (~6.5 ka BP) with a decrease in the GMST during the following six millennia prior to industrialization (Kaufman & Broadman, 2023).

6. Conclusions

Since 2002, unique tropical wetland vegetation preserved along the western margin of the QIC has been collected as the ice margin retreated and has been ¹⁴C-dated to document the ice cap's fluctuation during the mid-Holocene. Evidence from the North Lake adjacent to the current western ice margin indicates that the entire QIC behaves similarly to the North Lake Lobe in response to climatic fluctuations, with minor variations that depend on the topography of the bedrock and shape of each glacier in its respective valley. However, dating of vegetation in growth positions and other organic materials provides evidence from sites that experienced ice-free conditions along the retreating margins of the ice cap. Advances in the QIC over this time support other paleoclimatic studies which suggest a rapid climate transition during the mid-Holocene on local to global scales. From the extent of North Lake Lobe as well as other QIC outlet glaciers, there are indications that over the last 35 years the western margin of the Quelccaya ice cap has been retreating ~20 m/yr and the extent of the ice cap is now smaller than it has been in over 7,000 years.

Data Availability Statement

All locations and details of the mapped data are available in Supporting Information S1 along with outlet glaciers GEEiT margin outlines. The GEEiT outlines and ice extent isochrons are available from Lamantia et al. (2023) Zenodo at http://doi.org/10.5281/zenodo.10019398. The software used to monitor the recent retreat was the GEEDiT tool, which can be utilized at (https://code.earthengine.google.com/84ad6c8df0013331b-01ca991257dc73c) (Lea, 2018) and the Holocene ice movement contours and figure creation were completed via the QGIS software, accessed at (https://www.qgis.org/en/site/forusers/download.html). All location data for past integrated studies and this current study are available in Supporting Information S1.

References

Abbott, M. B., Seltzer, G. O., Kelts, K. R., & Southon, J. (1996). Holocene paleohydrology of the tropical Andes from lake records.

Anderson, R. K., Miller, G. H., Briner, J. P., Lifton, N. A., & DeVogel, S. B. (2008). A millennial perspective on Arctic warming from ¹⁴C in quartz and plants emerging from beneath ice caps. *Geophysical Research Letters*, *35*(1), L01502. https://doi.org/10.1029/2007GL032057

Baker, P. A., Fritz, S. C., Burns, S. J., Ekdahl, E., & Rigsby, C. A. (2009). The nature and origin of decadal to millennial scale climate variability in the southern tropics of South America: The Holocene record of Lago Umayo, Peru. In F. Vimeux, F. Sylvestre, & M. Khodri (Eds.), Past climate variability in South America and surrounding regions (Vol. 14, pp. 301–322). Springer Netherlands. https://doi.org/10.1007/978-90-481-2672-9_13

The previous field expeditions that collected the plant specimens were supported by several projects, including past grants from NSF, NOAA, and OSU We are appreciative of donations to the OSU Byrd Center Ice Core Salvage Fund GF 310873 that enabled the 14C dating of our samples. We are grateful to the researchers who contributed their data from previous studies, including Bryan Mark, Thomas Lowell, Meredith Kelly, Justin Stroup, and Aron Buffen. We thank Thomas Chudley for his assistance with the use of the GEEDIt tool to digitize the recent ice margins and Compton Tucker and Christopher Shuman for their assistance in acquiring the WV03 satellite image of the OIC. We acknowledge the many field team members who over the years have assisted in locating and sampling plant sites: Emilie Beaudon, Felix Benjamin, Giuliano Bertagna, Elizabeth Birkos, Aron M. Buffen, Paolo Gabrielli, Bradley Goodman, Aubrey Hillman, Katelyn Johnson, Natalie Kehrwald, Donald Kenny, Stanislav Kutuzov, Vicencio Maguiña, Vladimir Mikhalenko, Keith Mountain, Stacy Porter, Wilmer Sanchez Rodriquez, Roxana Sierra-Hernández David Urmann and Gustavo Valdivia Corrales. This is Byrd Polar and Climate Research Center contribution No. C-1622

9 of 11

- Baker, P. A., Fritz, S. C., Garland, J., & Ekdahl, E. (2005). Holocene hydrologic variation at Lake Titicaca, Bolivia/Peru, and its relationship to North Atlantic climate variation. *Journal of Quaternary Science*, 20(7–8), 655–662. https://doi.org/10.1002/jqs.987
- Baker, P. A., Seltzer, G. O., Fritz, S. C., Dunbar, R. B., Grove, M. J., Tapia, P. M., et al. (2001). The history of South American tropical precipitation for the past 25,000 years. *Science*, 291(5504), 640–643. https://doi.org/10.1126/science.291.5504.640
- Bar-Matthews, M., Ayalon, A., Kaufman, A., & Wasserburg, G. J. (1999). The Eastern Mediterranean paleoclimate as a reflection of regional events: Soreq cave, Israel. *Earth and Planetary Science Letters*, 166(1–2), 85–95. https://doi.org/10.1016/S0012-821X(98)00275-1
- Bond, G., Kromer, B., Beer, J., Muscheler, R., Evans, M. N., Showers, W., et al. (2001). Persistent solar influence on North Atlantic climate during the Holocene. *Science*, 294(5549), 2130–2136. https://doi.org/10.1126/science.1065680
- Bova, S., Rosenthal, Y., Liu, Z., Godad, S. P., & Yan, M. (2021). Seasonal origin of the thermal maxima at the Holocene and the last interglacial. *Nature*, 589(7843), 548–553. https://doi.org/10.1038/s41586-020-03155-x
- Bradley, R. S., Vuille, M., Diaz, H. F., & Vergara, W. (2006). Threats to water supplies in the tropical Andes. *Science*, *312*(5781), 1755–1756. https://doi.org/10.1126/science.1128087
- Brecher, H. H., & Thompson, L. G. (1993). Measurement of the retreat of Qori Kalis glacier in the tropical Andes of Peru by terrestrial photogrammetry. 6.
- Bronk Ramsey, C. (1995). Radiocarbon calibration and analysis of stratigraphy: The OxCal program. *Radiocarbon*, 37(2), 425–430. https://doi. org/10.1017/S0033822200030903
- Brook, E. J., & Buizert, C. (2018). Antarctic and global climate history viewed from ice cores. *Nature*, 558(7709), 200–208. https://doi. org/10.1038/s41586-018-0172-5
- Buffen, A. M., Thompson, L. G., Mosley-Thompson, E., & Huh, K. I. (2009). Recently exposed vegetation revealed Holocene changes in the extent of the Quelccaya Ice Cap, Peru. *Quaternary Research*, 72(2), 157–163. https://doi.org/10.1016/j.yqres.2009.02.007
- Chimborazo, O. (2022). Observations and simulated mechanisms of elevation-dependent warming over the tropical Andes. *Journal of Climate*, 35(3), 1021–1044. https://doi.org/10.1175/JCLI-D-21-0379.1
- Garreaud, R., Vuille, M., & Clement, A. C. (2003). The climate of the Altiplano: Observed current conditions and mechanisms of past changes. Palaeogeography, Palaeoclimatology, Palaeoecology, 194(1–3), 5–22. https://doi.org/10.1016/S0031-0182(03)00269-4
- Georges, C. (2004). 20th-century glacier fluctuations in the tropical Cordillera Blanca, Perú. Arctic Antarctic and Alpine Research, 36(1), 100–107. https://doi.org/10.1657/1523-0430(2004)036[0100:TGFITT]2.0.CO;2
- Gould, B. A., León, B., Buffen, A. M., & Thompson, L. G. (2010). Evidence of a high-Andean, mid-Holocene plant community: An ancient DNA analysis of glacially preserved remains. *American Journal of Botany*, 97(9), 1579–1584. https://doi.org/10.3732/ajb.1000058
- Hanshaw, M. N., & Bookhagen, B. (2014). Glacial areas, lake areas, and snow lines from 1975 to 2012: Status of the Cordillera Vilcanota, including the Quelccaya Ice Cap, northern central Andes, Peru. *The Cryosphere*, 8(2), 359–376. https://doi.org/10.5194/tc-8-359-2014
- Huber, U. M., & Markgraf, V. (2003). Holocene fire frequency and climate change at Rio Rubens Bog, southern Patagonia. In Fire and climatic change in temperate ecosystems of the western Americas (pp. 357–380). Springer.
- Hurley, J. V., Vuille, M., Hardy, D. R., Burns, S. J., & Thompson, L. G. (2015). Cold air incursions, δ¹⁸O variability, and monsoon dynamics associated with snow days at Quelccaya Ice Cap, Peru. *Journal of Geophysical Research: Atmospheres*, 120(15), 7467–7487. https://doi. org/10.1002/2015JD023323
- Kaser, G., & Osmaston, H. (2002). Tropical glaciers. Cambridge University Press.
- Kaufman, D., McKay, N., Routson, C., Erb, M., Dätwyler, C., Sommer, P. S., et al. (2020). Holocene global mean surface temperature, a multi-method reconstruction approach. *Scientific Data*, 7(1), 201. https://doi.org/10.1038/s41597-020-0530-7
- Kaufman, D. S., & Broadman, E. (2023). Revisiting the Holocene global temperature conundrum. *Nature*, 614(7948), 425–435. https://doi.org/10.1038/s41586-022-05536-w
- Kelly, M. A., Lowell, T. V., Applegate, P. J., Phillips, F. M., Schaefer, J. M., Smith, C. A., et al. (2015). A locally calibrated, late glacial ¹⁰Be production rate from a low-latitude, high-altitude site in the Peruvian Andes. *Quaternary Geochronology*, 26, 70–85. https://doi.org/10.1016/j. quageo.2013.10.007
- Kelly, M. A., Lowell, T. V., Applegate, P. J., Smith, C. A., Phillips, F. M., & Hudson, A. M. (2012). Late glacial fluctuations of Quelccaya Ice Cap, southeastern Peru. *Geology*, 40(11), 991–994. https://doi.org/10.1130/G33430.1
- Lamantia, K. (2018). Comparison of glacier loss on Qori Kalis, Peru and Mt. Kilimanjaor, Tanzania over the last decade using digital photogrammetry and stereo analysis. Ohio State University.
- Lamantia, K., Thompson, L. G., Mosley-Thompson, E., & Davis, M. (2023). Unique collections of ¹⁴C-dated vegetation reveal mid-Holocene fluctuations of the Quelccaya Ice Cap, Peru. https://doi.org/10.5281/zenodo.10019398
- Lea, J. M. (2018). Google Earth Engine Digitisation Tool (GEEDiT) and the Margin change Quantification Tool (MaQiT)-simple tools for the rapid mapping and quantification of changing Earth surface margins. *Earth Surface Dynamics*, 6(3), 551–561. https://doi.org/10.5194/ esurf-6-551-2018
- Liu, Z., Zhu, J., Rosenthal, Y., Zhang, X., Otto-Bliesner, B. L., Timmermann, A., et al. (2014). The Holocene temperature conundrum. Proceedings of the National Academy of Sciences of the United States of America, 111(34). https://doi.org/10.1073/pnas.1407229111
- López-Moreno, J. I., Fontaneda, S., Bazo, J., Revuelto, J., Azorin-Molina, C., Valero-Garcés, B., et al. (2014). Recent glacier retreat and climate trends in Cordillera Huaytapallana, Peru. *Global and Planetary Change*, 112, 1–11. https://doi.org/10.1016/j.gloplacha.2013.10.010
- Magny, M., Leuzinger, U., Bortenschlager, S., & Haas, J. N. (2006). Tripartite climate reversal in Central Europe 5600–5300 years ago. *Quaternary Research*, 65(1), 3–19. https://doi.org/10.1016/j.yqres.2005.06.009
- Marcott, S. A., Shakun, J. D., Clark, P. U., & Mix, A. C. (2013). A reconstruction of regional and global temperature for the past 11,300 years. *Science*, 339(6124), 1198–1201. https://doi.org/10.1126/science.1228026
- Mark, B. G., Seltzer, G. O., Rodbell, D. T., & Goodman, A. Y. (2002). Rates of deglaciation during the last glaciation and Holocene in the Cordillera Vilcanota-Quelccaya Ice Cap region, southeastern Perú. *Quaternary Research*, 57(3), 287–298. https://doi.org/10.1006/qres.2002.2320
- Mark, S. Z., Abbott, M. B., Rodbell, D. T., & Moy, C. M. (2022). XRF analysis of Laguna Pallcacocha sediments yields new insights into Holocene El Niño development. *Earth and Planetary Science Letters*, 593, 117657. https://doi.org/10.1016/j.epsl.2022.117657
- McCormac, F. G., Hogg, A. G., Blackwell, P. G., Buck, C. E., Higham, T. F. G., & Reimer, P. J. (2004). SHCal04 Southern Hemisphere calibration, 0–11.0 Cal Kyr BP. *Radiocarbon*, 46(3), 1087–1092. https://doi.org/10.1017/S0033822200033014
- Mercer, J. H., & Palacios, M. O. (1977). Radiocarbon dating of the last glaciation in Peru. *Geology*, 5(10), 600. https://doi. org/10.1130/0091-7613(1977)5<600:RDOTLG>2.0.CO;2
- Moy, C. M., Seltzer, G. O., Rodbell, D. T., & Anderson, D. M. (2002). Variability of El Niño/Southern Oscillation activity at millennial timescales during the Holocene epoch. *Nature*, 420(6912), 162–165. https://doi.org/10.1038/nature01194
- Munoz Uribe, P. A. (2012). Holocene climate variability in tropical South America: Case history from a high-mountain wet zone in NW Colombia based on palynology and X-ray microfluorescence. https://doi.org/10.13097/ARCHIVE-OUVERTE/UNIGE:34298



- Oppo, M., & Cullen, J. (2003). Deepwater variability in the Holocene Epoch. Nature, 422(277), 277. https://doi.org/10.1038/422277b
- Park, J., Park, J., Yi, S., Cheul Kim, J., Lee, E., & Choi, J. (2019). Abrupt Holocene climate shifts in coastal East Asia, including the 8.2 ka, 4.2 ka, and 2.8 ka BP events, and societal responses on the Korean Peninsula. *Scientific Reports*, 9(1), 10806. https://doi.org/10.1038/ s41598-019-47264-8
- Pepin (2015). Elevation-dependent warming in mountain regions of the world. Nature Climate Change, 5(5), 424–430. https://doi.org/10.1038/ nclimate2563
- Rodbell, D. T., Hatfield, R. G., Abbott, M. B., Chen, C. Y., Woods, A., Stoner, J. S., et al. (2022). 700,000 years of tropical Andean glaciation. *Nature*, 607(7918), 301–306. https://doi.org/10.1038/s41586-022-04873-0
- Rodbell, D. T., Seltzer, G. O., Anderson, D. M., Abbott, M. B., Enfield, D. B., & Newman, J. H. (1999). An~15,000-year record of El Niño-driven alluviation in southwestern Ecuador. Science, 283(5401), 516–520. https://doi.org/10.1126/science.283.5401.516
- Rodbell, D. T., Seltzer, G. O., Mark, B. G., Smith, J. A., & Abbott, M. B. (2008). Clastic sediment flux to tropical Andean lakes: Records of glaciation and soil erosion. *Quaternary Science Reviews*, 27(15–16), 1612–1626. https://doi.org/10.1016/j.quascirev.2008.06.004
- Schauwecker, S., Rohrer, M., Acuña, D., Cochachin, A., Dávila, L., Frey, H., et al. (2014). Climate trends and glacier retreat in the Cordillera Blanca, Peru, revisited. *Global and Planetary Change*, 119, 85–97. https://doi.org/10.1016/j.gloplacha.2014.05.005
- Seehaus, T., Malz, P., Sommer, C., Soruco, A., Rabatel, A., & Braun, M. (2020). Mass balance and area changes of glaciers in the Cordillera Real and Tres Cruces, Bolivia, between 2000 and 2016. *Journal of Glaciology*, 66(255), 124–136. https://doi.org/10.1017/jog.2019.94
- Soto Carrión, C., Zuñiga Negron, J. J., Paucar Ancco, J., Jiménez Mendoza, W., Ibarra Cabrera, M. J., Narváez Liceras, A., & Paucar Ancco, S. (2022). Multi-temporal analysis of the glacier retreat using Landsat satellite images in the Nevado of the Ampay National Sanctuary, Peru. Journal of Sustainable Development of Energy, Water and Environment Systems, 10(1), 1–15. https://doi.org/10.13044/j.sdewes.d8.0380
- Stroup, J. S., Kelly, M. A., Lowell, T. V., Applegate, P. J., & Howley, J. A. (2014). Late Holocene fluctuations of Qori Kalis outlet glacier, Quelccaya Ice Cap, Peruvian Andes. Geology, 42(4), 347–350. https://doi.org/10.1130/G35245.1
- Stroup, J. S., Kelly, M. A., Lowell, T. V., Smith, C. A., Beal, S. A., Landis, J. D., & Tapia, P. M. (2015). Late Holocene fluctuations of Quelccaya Ice Cap, Peru, registered by nearby lake sediments. *Journal of Quaternary Science*, 30(8), 830–840. https://doi.org/10.1002/jqs.2821
- Stuiver, M., & Reimer, P. J. (1993). Extended ¹⁴C data base and revised CALIB 3.0 ¹⁴C age calibration program. *Radiocarbon*, 35(1), 215–230. https://doi.org/10.1017/S0033822200013904
- Thompson, L. G. (2000). Ice core evidence for climate change in the Tropics: Implications for our future. *Quaternary Science Reviews*, 19(1–5), 19–35. https://doi.org/10.1016/S0277-3791(99)00052-9
- Thompson, L. G., Davis, M. E., Mosley-Thompson, E., Beaudon, E., Porter, S. E., Kutuzov, S., et al. (2017). Impacts of recent warming and the 2015/2016 El Niño on tropical Peruvian ice fields. *Journal of Geophysical Research: Atmospheres*, 122(23), 12688–12701. https://doi. org/10.1002/2017JD026592
- Thompson, L. G., Davis, M. E., Mosley-Thompson, E., Porter, S. E., Corrales, G. V., Shuman, C. A., & Tucker, C. J. (2021). The impacts of warming on rapidly retreating high-altitude, low-latitude glaciers and ice core-derived climate records. *Global and Planetary Change*, 203, 103538. https://doi.org/10.1016/j.gloplacha.2021.103538
- Thompson, L. G., Mosley-Thompson, E., Brecher, H., Davis, M., León, B., Les, D., et al. (2006). Abrupt tropical climate change: Past and present. Proceedings of the National Academy of Sciences of the United States of America, 103(28), 10536–10543. https://doi.org/10.1073/ pnas.0603900103
- Thompson, L. G., Mosley-Thompson, E., Davis, M. E., Henderson, K. A., Brecher, H. H., Zagorodnov, V. S., et al. (2002). Kilimanjaro ice core records: Evidence of Holocene climate change in tropical Africa. Science, 298(5593), 589–593. https://doi.org/10.1126/science.1073198
- Thompson, L. G., Mosley-Thompson, E., Davis, M. E., Lin, P.-N., Henderson, K. A., Cole-Dai, J., et al. (1995). Late glacial stage and Holocene tropical ice core records from Huascaran, Peru. Science, 269(5220), 46–50. https://doi.org/10.1126/science.269.5220.46
- Thompson, L. G., Mosley-Thompson, E., Davis, M. E., Zagorodnov, V. S., Howat, I. M., Mikhalenko, V. N., & Lin, P. N. (2013). Annually resolved ice core records of tropical climate variability over the past ~1800 years. *Science*, 340(6135), 945–950. https://doi.org/10.1126/ science.1234210
- Tiner, R. J., Negrini, R. M., Antinao, J. L., McDonald, E., & Maldonado, A. (2018). Geophysical and geochemical constraints on the age and paleoclimate implications of Holocene lacustrine cores from the Andes of central Chile: Holocene paleoclimate of the central Chilean Andes. *Journal of Quaternary Science*, 33(2), 150–165. https://doi.org/10.1002/jqs.3012
- Toledo, O., Palazzi, E., Cely Toro, I. M., & Mortarini, L. (2022). Comparison of elevation-dependent warming and its drivers in the tropical and subtropical Andes. *Climate Dynamics*, 58(11–12), 3057–3074. https://doi.org/10.1007/s00382-021-06081-4
- Urrego, D. H., Bush, M. B., & Silman, M. R. (2010). A long history of cloud and forest migration from Lake Consuelo, Peru. *Quaternary Research*, 73(2), 364–373. https://doi.org/10.1016/j.yqres.2009.10.005
- Vuille, M., & Bradley, R. S. (2000). Mean annual temperature trends and their vertical structure in the tropical Andes. Geophysical Research Letters, 27(23), 3885–3888. https://doi.org/10.1029/2000GL011871
- Vuille, M., Bradley, R. S., & Keimig, F. (2000). Interannual climate variability in the Central Andes and its relation to tropical Pacific and Atlantic forcing. *Journal of Geophysical Research*, 105(D10), 12447–12460. https://doi.org/10.1029/2000JD900134
- Vuille, M., Franquist, E., Garreaud, R., Lavado Casimiro, W. S., & Cáceres, B. (2015). Impact of the global warming hiatus on Andean temperature. Journal of Geophysical Research: Atmospheres, 120(9), 3745–3757. https://doi.org/10.1002/2015jd023126
- Weng, C., Bush, M. B., Curtis, J. H., Kolata, A. L., Dillehay, T. D., & Binford, M. W. (2006). Deglaciation and Holocene climate change in the western Peruvian Andes. *Quaternary Research*, 66(1), 87–96. https://doi.org/10.1016/j.yqres.2006.01.004
- Yarleque, C., Vuille, M., Hardy, D. R., Timm, O. E., De la Cruz, J., Ramos, H., & Rabatel, A. (2018). Projections of the future disappearance of the Quelccaya Ice Cap in the Central Andes. *Scientific Reports*, 8(1), 1–11. https://doi.org/10.1038/s41598-018-33698-z